Author's response to reviews

Title: Interdisciplinary team approach for complicated type B aortic dissection with concomitant hematotherax by Endovascular grafting and left side mini thoracotomy: a case report

Authors:

Marcel Vollroth (marcel-vollroth@gmx.de)
Jerog Seeburger (marcel-vollroth@gmx.de)
Philipp Kiefer (marcel-vollroth@gmx.de)
Michael Hoebartner (marcel-vollroth@gmx.de)
Yvonne Bausback (marcel-vollroth@gmx.de)
Jens Garbade (marcel-vollroth@gmx.de)
Friedrich W Mohr (marcel-vollroth@gmx.de)

Version: 2 Date: 28 August 2012

Author's response to reviews: see over
I declare, on behalf of ALL authors, that

1. The material in this article has not been previously published elsewhere in either print and/or electronic form and is not currently being considered for publication by any other journal (see: Eur J Cardiothorac Surg 1999; 16:1)

2. The work is the original creation of ALL authors cited only, there are no undisclosed authors (ghostwriters), and each author has read and approved the manuscript and agrees with its submission

3. ALL authors fulfill the criteria for authorship*

4. There is no undisclosed ethical problem or conflict of interest* and the research on which this article was based complies with all applicable guidelines in relation to legal and ethical practices

5. This paper has not been rejected or withdrawn from EJCTS or ICVTS previously (thus not eligible for re-submission)

6. The submission fully conforms to the journal’s instructions to authors** - the Editors reserve the right to REJECT OUTRIGHT any paper that does not

7. Written permission to reproduce any copyrighted material, for print and online publication in perpetuity, has been obtained and if necessary paid for by the author(s)

8. All authors agree that you will sign an exclusive license to publish on their behalf should the paper be accepted